Author's response to reviews

Title: Effect of Nigella sativa fixed and essential oils on antioxidant status, hepatic enzymes, and immunity in streptozotocin induced diabetes mellitus

Authors:

Muhammad Tauseef Sultan (tauseefsultan@hotmail.com)
Masood Sadiq Butt (drmsbutt@yahoo.com)
Roselina Karim (rosaz@putra.upm.edu.my)
Shahzad Zafar Iqbal (shahzad10542005@yahoo.com)
Shakeel Ahmad Ahmad (shakeel.agronomy@gmail.com)
Muhammad Zia-Ul-Haq (ahirzia@gmail.com)
Luigi Aliberti (luigialiberti83@libero.it)
Vincenzo De Feo (defeo@unisa.it)

Version: 2
Date: 16 December 2013

Author's response to reviews: see over
Dear Editor
BMC Complementary and Alternative Medicine

Subject: Submission of a manuscript

Dear Sir,
Please find attached revised paper entitled, "Effect of *Nigella sativa* fixed and essential oils on antioxidant status, hepatic enzymes, and immunity in streptozotocin induced diabetes mellitus" for submission in your esteemed journal.

1. We have added ethics statement.
2. We have moved table 1 at end.
3. We have improved English.

The paper has not been submitted nor under process anywhere else. Also please find names of potential referees.

**Dr. Milan S. Stanković**
Faculty of Science
University of Kragujevac
Street Radoja Domanovića No. 12, Kragujevac 34000, Serbia
Email. mstankovic@kg.ac.rs

**Dr. Diana Badiu**
Department of Natural Sciences
Faculty of Natural and Agricultural Sciences
Ovidius University from Constanta
1, University str., 2nd Campus
900470, Constanta- Romania
E-mail. gineco@pulsmedia.ro

**Dr. Imran Imran**
Campus Riedberg
Goethe University Frankfurt
Max-von-Laue Str. 9, 60438 Frankfurt
Email: imran@em.uni-frankfurt.de

Prof. Dr. Sagheer Ahmed
PAPRSB Institute of Health Sciences
University Brunei Darussalam
Email: sagheer.ahmed@ubd.edu.bn

Dr. Ehsan Elahi
Leipzig University
Rudolf Boehm Institute of Pharmacology and Toxicology
Härtelstrasse 16-18, 04107 Leipzig, Germany
E-Mail: elahi.ehsan@medizin.uni-leipzig.de

Kindly acknowledge receipt of this manuscript. I am available if there are any further queries.

-----

Highest Regards
Prof. Dr. Vincenzo De Feo
Dipartimento di Scienze Farmaceutiche e Biomediche,
Università degli Studi di Salerno, Italy
E-mail: defeo@unisa.it